Author’s response to reviews

Title: A Young Patient with visual loss

Authors:

Mark Taubert (mtaubert@doctors.org.uk)
Tim C Dowd (TIMDRED&AED@aol.com)
Angela Wood (Angela.Wood@stees.nhs.uk)

Version: 2 Date: 21 May 2007

Author’s response to reviews:

Please insert a Conclusions section after the Discussion. Done.
Please insert the Competing interests section after the Conclusions. If the authors have no competing interests, please state: "The authors declare that they have no competing interests." Done.
Please insert the Authors' contributions section after the Competing interests section. Done.
Please insert the author affiliations into the title page. Done.
Please insert the words "Case report:" into the title of the manuscript. Done.
Please upload the figures via the online submission system - they should not appear within the manuscript itself. I've taken them out completely.
The reference list should appear immediately after the Conclusions/Competing interests/Authors' contributions/Acknowledgements sections. Done.
Please confirm from whom the consent was granted - in this case the parent/guardian should have provided consent on behalf of the child. The patient in this case-report is 26 years old and thus able to give consent.